Author's response to reviews

Title: N-acetyltransferase 2 (NAT2) gene polymorphisms in colon and lung cancer patients

Authors:

Juergen Borlak (borlak@item.fraunhofer.de)
Stella Marie Reamon-Buettner (reamon-buettner@item.fraunhofer.de)

Version: 2 Date: 8 June 2006

Author's response to reviews: see over
Dear Editor:
Re: MS 1810758527102995, Borlak and Reamon-Buettner, *N-acetylimidazole 2 (NAT2)*
gene polymorphisms in colon and lung cancer patients

We were delighted to receive positive reviews on the above-named manuscript. We addressed the suggested minor revisions in Tables 1 and 2, and we revised the manuscript accordingly. Attached please find answer to the reviewer's comments.

We look forward to seeing our manuscript accepted for publication in BMC Medical Genetics.

Sincerely yours,

Juergen Borlak
Answer to Reviewers:

Reviewer 1 (Sara Raimondi)

1. Tables 1 and 2: The ORs cannot be calculated when one cell in the 2x2 table is empty, i.e. when there are no cases and controls with a certain genotype. For this reason, the authors should delete ORs for genotypes CC/AA/GA, CC/GA/GA (TABLE 1) and CC/GG/GA, CC/AA/GG, CC/AA/GA (TABLE 2).

As requested, we deleted the ORs for the above genotypes and revised the manuscript accordingly.